Author's response to reviews

Title: ADHD in adolescents with borderline personality disorder

Authors:

- Mario Speranza (msperanza@ch-versailles.fr)
- Anne Revah-Levy (anne.revah-levy@ch-argenteuil.fr)
- Samuele Cortese (samuele.cortese@gmail.com)
- Bruno Falissard (falissard_b@wanadoo.fr)
- Alexandra Pham-Scottez (a.pham@ch-sainte-anne.fr)
- Maurice Corcos (maurice.corcos@imm.fr)

Version: 4 Date: 28 September 2011

Author's response to reviews:

To the editorial team

BMC Psychiatry

As asked, we made the last editorial modifications of the paper “ADHD in adolescents with borderline personality disorder” (manuscript N°1660765808570547) to be published in BMC Psychiatry.

We hope that these modifications will satisfy the journal editorial style

We look forward to reading from you.

Sincerely yours,

Signature of the corresponding author

Mario Speranza MD PhD